Author’s response to reviews

Title: Virological failure reduced with HIV-serostatus disclosure, extra baseline weight and rising CD4 cells among HIV-positive adults in Northwestern Uganda

Authors:

Jonathan Izudi (jonahzd@gmail.com)
Sunday Alioni (alionisunday@yahoo.com)
Emmanuel Kerukadho (kerukadho@yahoo.co.uk)
David Nduguntse (davidndungutse@yahoo.co.uk)

Version: 1 Date: 28 Apr 2016

Author’s response to reviews:

The Editor’s concerned was, “The study presented by the authors has overall a small number of patients with a relatively short follow up. The data about disclosure and virological failure seem to me the most interesting finding. Since this cohort is still valuable, I suggest that the authors wait to have a larger samples size and a longer follow up and focus on the disclosure data, considering including also patients on other regimens”.

We have included data on patients on other ART regimens. The sample size is 383 instead of 248. The title was as well revised